Author's response to reviews

Title: Increased plasma soluble endoglin levels as an indicator of cardiovascular alterations in hypertensive and diabetic patients

Authors:

Ana M. Blázquez-Medela (grym@usal.es)
Luis García-Ortiz (lgarciao@usal.es)
Manuel A. Gómez-Marcos (magomez@usal.es)
José I. Recio-Rodríguez (donrecio@gmail.com)
Ángel Sánchez-Rodríguez (asanro@usal.es)
José M. López-Novoa (jmlnovoa@usal.es)
Carlos Martinez-Salgado (carlosms@usal.es)

Version: 3 Date: 22 November 2010

Author's response to reviews:

Dear Sirs,

Thank you very much for your positive evaluation and the acceptance of our manuscript in BMC Medicine

As suggested by the Editorial Production Team, we (the authors) want the tables to be visible within the final published manuscript. Thus, we have include them in the manuscript in a tables section following the references.

Looking forward to hear from you soon

Carlos Martinez-Salgado